Author's response to reviews

Title: Assessment of the medicinal potentials of the methanol extracts of the leaves and stems of Buddleja saligna.

Authors:

Adeolu A Adedapo (adedapo3a@yahoo.co.uk)
Florence O Jimoh (asj02@yahoo.com)
Srinivas Vedic (sreenivasvedicum@gmail.com)
Patrick J Masika (pmasika@ufh.ac.za)
Anthony J Afolayan (aafolayan@ufh.ac.za)

Version: 2 Date: 23 April 2009

Author's response to reviews:

The Editor,
I will be very pleased if the revised manuscript is given adequate attention and thereafter accept it for publication.

Adeolu Adedapo
Corresponding author.